Author's response to reviews

Title: Effect of 50% ethanolic extract of Syzygium aromaticum (L.) Merr. & Perry. (clove) on sexual behaviour of normal male rats

Authors:

Tajuddin D . . (tajuddin_amu@rediffmail.com)
Shamshad D Ahmad . (shamshad_2003@rediffmail.com)
Abdul D Latif . (latifabdul@rediffmail.com)
Iqbal D Ahmad Qasmi (qasimiqbal@rediffmail.com)

Version: 8 Date: 3 November 2004

Author's response to reviews:

MS:3430433273709239
Article title: Effect of 50% ethanolic extract of Syzygium aromaticum (L.) Merr. & Perry.(clove) on sexual behaviour of normal male rats.
Authors:Taajuddin, Shamshad Ahmad, Abdul Latif and Iqbal Ahmad Qasmi.

Dear Editorial Administrator
BMC, Journals
Thanks for informing about the acceptance of my paper for publication in your journal "BMC Complementary and Alternative Medicine ".
I am sending herewith appoint by-point detail of the formatting changes made in the manuscript as required and suggested for your consideration.

Formatting changes
Title :
The word "Sexual" has now been changed to 'sexual'
Author list:
The list of authors in the manuscript has now been written exactly as t hey are in the submission system.
The authors' names have been written identically and in the order in which they are listed in the submission system.
Abstract:
The colons from the abstract subheading in the manuscript file have been removed. But this removal could not make in submission system because in the submission system the Abstract of revise manuscript existed.
Materials & Methods:
This section has now been renamed as 'Methods'.
Section headings:
Italicisation of the section heading have been removed.
Statistical Analysis:
This has been changed to 'Statistical analysis'.
Author's contributions:
This has been changed to 'Authors' contributions.
Competing interests :
A' Competing interests' section between the Conclusions and Authors 'contributions sections has been included and explained as' The authors declare that thy have no competing interests.
Page 14:
The information between the reference sections and Table 1 has been deleted.
Tables:
All tables have now been reoriented in portrait form.
Table titles:
Each table title has now been placed above the tables. The sentence" All authors read and approved the final manuscripts" has been added in Authors' contributions' section.
Tables:
All tables have now been formatted as suggested.